Author's response to reviews

Title: Antiphospholipid syndrome nephropathy - a diagnostic clue to primary antiphospholipid syndrome masquerading as vasculitis: a case report

Authors:

R. A. Abeysekera (rajithaasa55@gmail.com)
A.W.M. Wazil (awmwazil@hotmail.com)
N. Nanayakkara (nishansrikandy@yahoo.co.in)
N. V. I. Ratnatunga (neela72002@yahoo.com)
K.H.M. Fernando (maithri2000@gmail.com)
J. Thinnarachchi (jalitha@ymail.com)

Version: 4 Date: 15 December 2014

Author's response to reviews:

The Editors
The Journal of Medical Case Reports
Re: Journal of Medical Case Reports MS: 3823048491409980

Title:
Antiphospholipid syndrome nephropathy – a diagnostic clue to primary antiphospholipid syndrome masquerading as vasculitis: a case report

As requested by the associate editor the following revisions have been done to the manuscript.

The authors need to remove 'unmarried' from the case report, as I do not think it is relevant to the case or its contents."

The word "unmarried" has been removed as requested

Thanking You,

Sincerely,

Dr.R.A.Abeysekera
Corresponding author